Author's response to reviews

Title: Spontaneous rupture of the long head of the biceps tendon in a hypothyroid female patient: A case report

Authors:

Konstantinos Pantazis (pantazis.kon@gmail.com)
Dimitrios N Roupas (nikosroupas@hotmail.com)
Andreas Panagopoulos (andpan21@gmail.com)
Stauroula Theodoraki (roula-theodor@hotmail.com)
Aggeliki Tsintoni (agtsintoni@gmail.com)
Venetsana Kyriazopoulou (vkyriazopoulou@med.upatras.gr)

Version: 4 Date: 8 May 2015

Author's response to reviews:

we have restructured the abstract according to your guidelines, adding our impression that this a unique report not previously published
we made the corrections in the title page
we changed the gender to female